

ST. MARY'S UNIVERSITY SCHOOL OF GRADUATE STUDIES

A THESIS ON

FCTORS AFFECTING EMPLOYEE TURNOVER INTENTION: IN CASE OF NIB INTERNATIONAL BANK S.C (NIB)

BY: RAHEL YUKUNOAMLAK

ADVISOR: MATIWOS ENSERMU (PhD)

JUNE, 2015 ADDIS ABABA, ETHIOPIA

I A THESIS ON

FCTORS AFFECTING EMPLOYEE TURNOVER INTENTION IN CASE OF NIB INTERNATIONAL BANK S.C (NIB)

A THESIS SUBMITTED TO ST.MARY'S UNIVERSITY, SCHOOL OF GRADUATE STUDIES IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF MASTER OF BUSINESS ADMINISTRATION

BY: RAHEL YUKUNOAMLAK

ENROLMENT No: SGS/0232/2005

ADVISOR: MATIWOS ENSERMU (PhD)

JUNE, 2015 ADDIS ABABA, ETHIOPIA

SCHOOL OF GRADUATE STUDIES FACULTY OF BUSINESS

A THESIS ON FCTORS AFFECTING EMPLOYEE TURNOVER INTENTION IN CASE OF NIB INTERNATIONAL BANK S.C (NIB)

SUBMITTED BY:

RAHEL YUKUNOAMALK

APPROVED BY BOARD OF EXAMINERS

Dean, Graduate Studies	Signature
Advisor	Signature
External Examiner	Signature
Internal Examiner	Signature

Ш

DECLARATION

I the undersigned, declare that this thesis is my original work, prepared under the guidance of Dr. Matiwos Ensermu. All sources of materials used for the thesis have been duly acknowledged. I further confirm that the thesis has not been submitted either in part or in full to any other higher learning institution for the

purpose of earning any degree.			
Name Signature			
St. Mary's University, Addis	Ababa	June, 2015	
	IV		
1	ENDORSEMI	ENT	
This thesis has been submitted. Studies for examination with m			ol of Graduate
	Advisor Name		
A	dvisor Signatuı	re	
St. Mary's University, Addis	Ababa June, 2	2015	

ABSTRACT

The present research attempts to examine the relationship with Turnover Intention and organizational factors. A cross-sectional survey was conducted on a random sample of employees from twenty one branches of Nib International Bank, a private bank in Ethiopia. Correlation and regression analyses revealed a number of significant relationships between the two variables. The findings of the study found dissatisfaction with benefits and reward system, the poor job and organizational specific communication, dissatisfaction with the work itself are the most important organizational factors to cause turnover intention of employees.

Key Words: Job satisfaction, Turnover Intentions, Pay, Nature of work, Supervision

VI

ACKNOWLEDGEMENT

I thank the Almighty God for His Graces, which enabled me to complete this work. In a special way, I

express my sincere thanks and gratitude to Dr. Matiwos Ensermu (Supervisor) for his direction,

corrections and suggestions particularly on how I should undertake this work and his time spent in

reviewing my work.

I extend my gratitude to my colleagues for their cooperation, encouragement, constructive criticisms,

intellectual support and guidance. Accomplishment of this work would have been impossible in the

absence of data; many thanks are due to all respondents from Nib International Bank S.C their kind

cooperation towards successful completion of this work. Be blessed always.

I am greatly indebted to my former colleagues Habtamu Taye. I am very grateful for your

unconditional support in every aspect and for his inspiration to start education and for his valuable

comments. My heartfelt appreciation goes to Daniel Kemaw for his unreserved effort in providing

all the required assistance and understanding. Furthermore I would like to thank my mother, Sister

and Brothers for their endless love and support.

Finally, I want to express my deepest thanks those who encouraged me, which I did not mentioned

their name, by providing all the necessary support in my endeavor.

VII

LIST OF ABBREVIATIONS

NIB: Nib International Bank

ATM: Automated Transfer Money

PoS: Point of Sales

TABLE OF CONTENTS

COVARE PAGE	Page I
BOARD OF EXAMINER	
BOARD OF EAAMINER	111
DECLARATION	3
ABSTRACT	5I
ACKNOWLEDGEMENT	
\mathbf{V}	
LIST OF ABBREVIATIONS	
TABLE OF CONTENTS	
CHAPTED ONE	<u>Xi</u>
CHAPTER ONE	
INTRODUCTION	
1.1. Background of the Study	
1.2. Background of the Organization Study	
1.3. Statement of the Problem	
1.4. Objective of the Study	
1.4.1. General Objective	6
1.4.2. Specific Objectives	6
1.5. Research Questions	6
1.6. Significance of the Study	7
1.7. Delimitation/Scope of the Study	7
1.8. Organization of the Report	7
CHAPTER TWO.	
REVIEW of RELATED LITERATURE	8
2.1. Definition of Employee Turnover	
2.2. Types of Turnover	
2.2.1 Voluntary	
Turnover	
8	
2.2.2 Involuntary Turnover	
2.2.3 Functional vs Dysfunctional Turnover	
2.3.1. Demographic Factors	11
2.3.2. Personal Factors	12
2.3.3 Pull Factors	12

2.3.4. Pus	h Factor	12
2.3.5. Org	anizational Factors	13
2.3.6.Atti	tude Factors	13
2.4. Effects	of Employees Turnover	14
	ds of Identifyning Reasons for staff Turnover	
	rview	
	veys of EX-	
		•••••
15		
2.5.3. Attit	aude	
veys		•••••
2.5.4. Qua		
		•••••
16 2.6. Reducii	ng Labour Turnover	
16		
	pensation and	
Benents 16		• • • • • • • • • • • • • • • • • • • •
-	ognition and	
Rewards	······································	•••••
. 18	wing Dueforsional Development and Course Dlaming	
2.0.3. 1 rai	ning, Professional Development and Career Planning18	
2.6.4. Recr	uitment and	
		20
	thy Workplace and Well being	
2.6.6. Worl		
Balance		
2		
2.6.7. Job 1 Teams	Design and Work	22
	R THREE	
_	CSEARCH DESIGN AND METHODOLOGY	2
3.1.1.	Research Design	
	pulation and Sampling Techniques	
<u>3.2.1.</u>	Population	
<u>3.2.2.</u>	Sampling Techniques	23
<u>3.2.3.</u>	Sample Determination	23
3.3. Re	search data sources.	24
<u>3.3.1.</u>	Primary data research	24
3.3.2.	Secondary data research	24
3.3.3.	Research questionnaires	
3.3.4.	Procedures of Data Collection	
.J.,J.4.		

<u>3.3.</u>	3.5. Analysis of data	26
3.3.	3.6. Descriptive Analysis	27
<u>3.3.</u>	8.7. Research limitations	27
<u>3.4.</u>	Time restraints.	27
<u>3.5.</u>	Access to key information and information holders	27
<u>3.6.</u>	Study Model	27
CHAP	PTER FOUR	30
	Presentation and	
	is	30
	. Surveys	
Results	S	•••••
	escriptive Analysis in terms of frequency per each	
	le31	
	escriptive	
Statisti	ics	•••••
	sults for Pearson correlation Coefficient	
	40	
	sults of the Regression is	41
4.6. Disc		, 41
		•••••
	44 Supervision Support	
	······································	44
	Benefits and ls	
45		,
	Nature of the	
Work 45		• • • • • • • • • • • • • • • • • • • •
	Organization and Job Specific communication	
•••••	46	
CHAP	PTER FIVE	44
	mary of Findings, Conclusion and	_
Recomi	mendation47	
5.1	1. Summary of	
tinding 47	gs	•••••
5.2.		
Conclu	ısion	
	48	
	commendation	
40		• • • • • • • • •

ELECTRONIC	
REFERENSES.	52
APPENDEX	
LIST OF TABLES	
Table 1 Sample Determination	25
Table 2	27
Table 4.1.1 Sample Size	
Table 4.1.2 Respondents Educational Background	
Table 4.1.3 Length of Stay in Current Bank of the Respondents in Years	
Table 4.2.1 Fair Pay	
Table 4.2.2 Appreciation	
Table 4.2.3 Promotion	
Table 4.2.4 Competent Supervision	
Table 4.2.5 Little interest for Subordinates	
Table 4.2.6 Easy Communication with Supervisor	34
Table 4.2.7 Rules and Procedures	
Table 4.2.8 Too Much Work	
Table 4.2.9 Incompetence of People	
Table 4.2.10 Fitting at Work	
Table 4.2.11 Clear Responsibilities	
Table 4.2.12 Enough Time	
Table 4.2.14 Clear Organizational Goals	
Table 4.3.1 Descriptive Statistics	
Table 4.4.1 Correlation Matrix	
Table 4.5.1 Variables Entered/Removed	
Table 4.5.2 Model Summary	
Table 4.5.3 ANOVA	
Table 151 Coefficients	

CHAPTER ONE

1. INTRODUCTION

1.1. Background of the Study

In this world of globalization there are growing changes and challenges in the society in which we leave and in which our work organizations operates. There are also growing changes and challenges taking place inside work organizations themselves. These changes and challenges are having an impact upon the management of people in work organization and the performance of these organizations on the one hand and upon the environment in which work organization must exist on the other hand. (Ngirwa, 2005). Employee turnover is one of the challenges that faced the work organization and have impact in the performance of those organizations.

Employee turnover is a persistent phenomenon that produces a serious problem for organizations. The loss of qualified and skilled employees is related to the reduction in the competitiveness, innovation, and quality of offered service to customers (Miller, 2010). Moreover, turnover entails significant costs of recruitment, selection, hiring and training of newcomers. Turnover of employees can be voluntary, involuntary, functional or dysfunctional.

Turnover can be described as a process in which job dissatisfaction is the first step, followed by intention to leave, which finally, can result in actual turnover (Bannister and Griffith, 1986). According to Meyer (1993) turnover intention is a conscious willfulness to seek other alternative job opportunities in other organization. Intentions are, according to researchers such as Ajzen and Fishbein (1980) and Igbaria and Greenhaus (1992), the most immediate determinants of actual Behavior.

Recent research has shown the negative effects of turnover alone on performance, organizational deviance, and organizational citizenship behaviors (Krishnan & Singh, 2010). Practically, most employees make their decision to leave over a lengthy period lasting months or even years. This

means organizations have time to implement positive changes and retain workers that have intentions to leave before it is too late or actual turnover happens.

There are number of factors that contribute to employee turnover. One of the most common reasons given for leaving the job is the availability of higher paying jobs. In a better economy the availability of alternative jobs plays a role in turnover, but this tends to be overstated in exit interviews.

Many academic scholars pointed out turnover as the correlation between job dissatisfaction and decreased organizational commitment. Karsh, Booske&Sainfort (2005) suggested that job satisfaction and organizational commitment are predictors of turnover intention. According to Guimaraes&Igbaria (1992) organizational commitment was an intervening variable in models of turnover and job satisfaction had direct effect on turnover intentions, and indirect effect through organizational commitment (Aryee, Wyatt & Min, 1990) have also identified job satisfaction and organizational commitment as main predictors of turnover intention. A research in Turkish context by Wasti (2003) also proved organizational commitment to be a predictor of turnover intentions. The stronger the feeling of discontent in one's job, the more likely one is to begin a search for an alternative job (Karsh et al., 2005). This perspective on turnover highlights the role of emotions and attitudes as antecedents of behavior. Spector (1997), the proposed model specifies how each organizational factors associated to Job satisfaction independently contribute to the prediction of turnover intention.

The arrow shows the predicted dependent-independent relationship among the variables Spector (1997) specifies that organizational factors like pay, promotion opportunity, supervision, coworkers, nature of the work, operating procedures, personal development, fringe benefits, Performance management, contingent rewards and communication associated to job satisfaction independently contribute to turnover.

Therefore this study tries to focus on ascertaining how employees' satisfaction with variables such as Pay, Promotion opportunity, Supervision, Co-workers, Nature of the work, Operating procedures, Personal development, Fringe benefits, Performance management, Contingent rewards and Communication affect turnover of employees and to measure the extent of the

influence of these different organizational factors on turnover of employees by testing formulated hypotheses so as to recommend the possible solution in order to tackle these most critical factors in one of the private banks, Nib International Bank.

1.2. Background of the Organization Study

As per the manual of Nib International Bank S.C. (NIB), it is one of the private banks in Ethiopia, which was established on 26 May 1999 under License No. LBB/007/99 and commenced operation on 28 October 1999, with a paid up capital of Birr 27.6 million and authorized capital Birr 150 million. The number of shareholders and employees were 717 and 27, respectively. It joined the banking industry as the sixth private bank licensed in the country. According to the annual report of the Bank, NIB registered a steady and remarkable growth and achievement in all aspects of its operations since its establishment. At the end of 2013/14 financial year, its total deposits stood at birr 7.9 billion. Total assets also grew to Birr 10.7 billion and paid up capital, reserves, retained earnings and undistributed profit reached Birr 1.9 billion. In terms of paid up capital the Bank is one of the strongest banks in the country.

NIB has been profitable since its first year of operation. The level of net profit before tax and after tax which were only Birr 18.0 million and Birr 11.6 million in 2001/2002, stood at over Birr 314 million in 2013/14 financial year.

The Bank's shareholders now reached 3894 and its customer's base exceeds more than 200,000. This clearly demonstrates the high level esteem it enjoys from the public.

Currently, the bank's employees reached about 2582. It has 104 branches, of which 62 are located in Addis Ababa and the remaining in regional towns.

In order to make more efficient its service delivery, the Bank has replaced core banking system with the most up-to-date system called T -24 since August 2013; a banking software of Temenos and electronic Payment Service ATM service with other five private banks currently together by the coordination of PSS (Premium Switch Solution). Besides, it has introduced electronic payment service (ATM, PoS, Mobile banking etc) together with Awash International Bank and United Bank.

1.3. Statement of the problem

Employee turnover has been recognized as a major managerial concern in contemporary work organizations. First, replacing employees may be costly, both in recruiting and training employees to obtain satisfactory levels of performance over time (Collins and Smith, 2006). In addition, high levels of employee turnover may impede the quality, consistency and stability of services that organizations provide to clients and customers (Trevor and Nyberg, 2008) and in turn increased client dissatisfaction with the services provided by the organization (Lin and Chang, 2005).

In banking organization the Department of Human Resource Management has become a focal point for organization performance. It is in this department where by planning, recruitment, motivation, compensation and training need assessment is done.

Employees in banking organizations are crucial in the building of service excellence as there is high competition (Svensson and Svaeri, 2011). A critical issue in service organizations may be to retain service employees in general, and specifically those employees who are talented in working with customers and delivering excellent service quality (Svensson and Svaeri, 2011).

Many reasons explain why employees withdraw from an organization especially in an economy where skills are relatively scarce and recruitment is costly, or where it takes several months to fill vacant positions as the present situation in the Ethiopian public sector.

Employees voluntarily resign their appointments in organizations for various reasons which can be classified into two: pull and push factors. The pull factors, according to Sherratt (2000:38) include the attraction of a new job especially in a growing economy. In such cases, it is the availability of alternative jobs that attracts an employee to withdraw from a particular organization Sherratt (2000:38) also explains that the push factor may be dissatisfaction with the present job that motivates an employee to seek alternative employment elsewhere. Sometimes, it is a mixture of both the pull and push factors. However, some reasons for leaving are entirely explained by domestic circumstances outside the control of any employer, as is the case when employees relocate with their spouses or partners.

Recent research by the British Chartered Institute of Personnel and Development Shows that push factors are a great deal more significant in most resignations than most managers appreciate. The research contends that it is relatively rare for people to leave jobs in which they are happy, even when offered higher pay elsewhere. Research conducted by the Hay group and reported by Sharman et al. (2006:22) reveals that about one third of the millions of employees surveyed worldwide plan to quit their jobs within two years.

Various studies were carried out to understand the major causes of employees turnover and retentions mechanisms that organizations should develop. Most of these studies were carried out by developed countries and few developing countries. Some studies indicate that every aspect of organizations is employees because turnover of employee leads to incurring of costs.

The banking industry in Ethiopia has witnessed a remarkable growth over the past two decades. According to the annual report of the NIB, the establishment of private banks and the resultant improvement in financial intermediation has been at the forefront for the commendable achievement registered so far. The report also states that the industry has been one of the most lucrative sectors in the country fetching an average return on equity every year. It can be said that private banks have been in honeymoon until recently. Nib International Bank has also enjoyed similar success stories.

However, Nib International Bank is currently under a serious threat from high rate of employee turnover. The turnover has been at an alarming rate over the past 2 years. The official data that has indicated the exit of 192 employees before nine months. Biannual Report of NIB calls the attention of both the management and any one that comes across of the information. Since July 2013, eight percent of the employees have left the organization.

Therefore, the purpose of this study is to assess employees' turnover in Nib International Bank and to investigate the intention and causes of employee turnover as well as proposing employee retention strategies.

And this study aims at investigating and finding solutions to the question "why is the employee turnover rate at the increase" given all the measures and efforts the Nib International Bank S.c has taken to address the problem? This is because many studies was conducted as will be shown

in the literature review but still there is a research gap on the retention program that organizations adopts in solving this problem. Perhaps this study will help to identify the solution to solve or reduce the gap.

1.4. Objective of the study

The study has the following general and specific Objectives.

1.4.1. General Objective

The general objective of the study is to assess the factors affecting of employees' turnover intention in NIB.

1.4.2. Specific Objectives

Based on the above general objective, the study would have the following specific objectives. These are:-

- To identify the factors which lead to employee turnover in NIB.
- To examine the different demographic factors that affect employee turnover in NIB
- To investigate whether employees are satisfied with the work if dissatisfied to identify the reasons.
- To identify the effectiveness of retention strategies that needs to be in place.
- To find out the effect of employees' perception on the application of organizational procedures on turnover intention.

1.5. Research Questions

The following research questions are designed to guide the attainment of the objectives of the study.

- What are the factors which lead to employee turnover in NIB?
- Is there any retention strategies adopted by the organization? If yes are the retention strategies effective?
- What is the importance of having a retention strategy to the NIB and what are the variables that the retention strategy should focus?

1.6. Significance of the Study

This study is expected to generate the following three benefits. The result of this study has greater importance for those who are responsible for the designing and administering employees' retention and development system of NIB to achieve organization objective through managing the behavior of individual employees. It also helps other similar service giving private banks in Ethiopia.

The findings and recommendations of this research bridges the gap between the knowledge, rationales, existing systems and functions nationally and the prevailing practices and experiences in NIB in terms of handling employees' behavior.

Since the research work is carried out to partially fulfill the requirements for the award of the Masters of Arts in Human Resource Management, it provides personal development and learning opportunities to the researcher.

1.7. Delimitation/Scope of the Study

The researcher is expected to face a time constraints since this is an academic paper and needs to be completed according to the academic calendar, hence it will be a short period of time for making the whole research report compiled, reviewed and analyzed thoroughly.

The researcher could be willing to cover more than one private bank; however it won't be possible due to financial constraints or budget constraints.

1.8. Organization of the Report

This study is organized in five chapters as follows. Chapter one includes the introduction which encompasses the background of the study, statement of the problem, basic research questions, objectives of the study, hypothesis, significance of the study and delimitation /scope of the study. Chapter two explores the literature review related to factors of job and turnover intention of employees. Chapter three explains the research design and methodology employed. In chapter four the researcher summarizes the results and interprets and discusses the findings. The last chapter, chapter five concludes the paper by providing summary and recommendations.

CHAPTER TWO

REVIEW of RELATED LITERATURE

Employee turnover is a major concern for organizations since it leads to increased hiring and training costs; loss of continuity and more importantly affects organizational productivity (Firth, Mellor, Moore, Loquet, C., 2004). These important consequences of turnover, have led to much research of this topic. However, the lack of standard records of why employees leave Lee and Mitchell (1994) and Morrel (2004) have stymied research in this area. Organizations need to understand the reasons why employees leave in order to lower their turnover rates.

2.1 Definition of Employee Turnover

Employee turnover (sometimes known as labour turnover or wastage or attrition) is the rate at which people leave an organization (Armstrong, 2010).

Stan Kossen (1991) defined staff turnover as it is the amount of movement in and out (of Employees) in an organization.

Employee turnover is defined as the intention to exit an organization voluntarily (Nazim, 2008). Turnover intention or Intent to leave the organization is one's behavioral attitude to withdraw from the organization, while turnover is considered the actual separation from the organization. Intent to leave has been identified to be the strongest predictor of actual turnover in the studies that have actually examined turnover (Keaveney, 1992). Intent to leave is based upon an attitudinal variable and is most typically found in job-related turnover (Cotton and Tuttle, 1986).

2.2 Types of Turnover

Two popular ways of classifying employee turnover are voluntary versus involuntary and functional versus dysfunctional.

2.2.1Voluntary Turnover

The specific form of turnover of interest for most organizational research is voluntary separations or quits (Bluedorn, 1982); (Price, 1997).

According to Lee & Maurer (1997) voluntary employee turnover has been a growing concern amongst HRM managers and researchers during the last century (Barnard, 1938); March & Simon (1958) probably because retention of knowledge workers can become a source of competitive advantage for a company (Sigler, 1999).

Voluntary turnover refers to an employee's chosen exit from the organization. They may be accepting employment with another company, relocating to a new area or dealing with a personal matter that makes it impossible to work. (http://smallbusiness.chron.com)

Theoretical Base for Voluntary Employee Turnover:-

Research and theory regarding voluntary employee turnover have for a long time been mainly influenced by 'Perceived ease and desirability of movement' framework (Lee & Mitchell, 1996). The perceived ease of movement has over the years evolved to mean perceived job alternatives Lee & Mitchell (1994) whilst the perceived desirability of movement is represented by job satisfaction (Jackofsky & Peters, 1983).

A dissatisfied employee undergoes in deciding when to terminate his/her employment with the organization (Mobley, 1977). Here it can be seen that voluntary turnover is based on the personal individual decision of an employee to resign to work for a particular organization.

2.2.2 Involuntary Turnover

These type of turnover usually initiated by the employer or they occur when management decides that it needs to terminate its relationship with an employee due to an economic necessity or poor fit between employee and the organization. It is the result of very serious and painful decisions that can profound impact on the entire organization, especially the employee loosing his/her job.

There are two types of involuntary separations; Discharge and Layoff. Discharge- takes place when management decides that there is a poor fit between an employee and the organization.

The discharge is a result of poor performance or failure to change some unacceptable behavior that management has tried repeatedly to correct. Layoff, employees lose their jobs because a change in a company's environment or strategy forces it to reduce the size of its work force. Global competition, reduction in product demand, changing technologies are those reduce the need for workers. Layoff has a powerful impact on the organization's remaining employees, who fear losing their job in the future.

2.2.3 Functional vs. Dysfunctional Turnover

Functional Turnover describes the situation where a poorly performing employee leaves the organization). It is, therefore, functional from the employer's perspective. Dysfunctional Turnover on the other hand is a term used to describe the situation when high performing employees leave the organization which is costly to the organization and considered a problem when it constitutes of several of such kind of employees. (http://www.mintrac.com.au)

Turnover Causes and Influencing Factors

According to the study by Bowen & Sihel (1997) turnover is a consequence of two variables. The first one is; (1) Perceived desire of movement, whereat high desire is due to lack of job satisfaction and lack of internal careers opportunities. The other variable, (2) Perceived ease of movement, are due to the number of external job alternatives. These variables can help organizations to predict voluntary turnover.

Researchers like Teeraprasert, Piriyakul and Khantanapha (2012) find out that employees who leave their organization are highly affected by the lack of job satisfaction. Their study focuses on voluntary turnover on dissatisfaction. According to the study the decision to leave is often associated with employees being unhappy with their current job and comes across another more attractive opportunity. Companies try to fight dissatisfaction in hope of reducing loss of valuable staff, as many studies would recommend (Lee & Mitchell 1994); (Bowen & Sihel, 1997); (Lee & Mitchell, 2004). Fredric Herzberg and his associates conducted a famous survey of 200 accountants and engineers that laid the foundation for the motivator-hygiene theory from 1959 (Ramlall, 2004). Herzberg's theory indicates that eliminating factors contributing to dissatisfaction does not necessarily increase job satisfaction. Consequently, factors that can create dissatisfaction must be met before factors affecting job satisfaction can be activated (ibid). Ronra and Chaisawat (2009) conducted their study of employee turnover based on Herzberg's' motivator—hygiene theory. The result of the findings indicate that factors affecting employees' turnover was inadequate recognition from managers, inadequate salary and benefits that did not meet the employees' needs, salary and responsibilities were not compatible and inadequate career path advancement. In contrast, the factors which influenced employees' job satisfaction were that the employee felt appreciated for completed task and that the company recognized their achievements. The pride to work for the organization, the understanding of how the job aligns with the company's mission and the balance between job and families responsibilities were also seen as important for the employees (Ronra et al., 2009).

Another research studies indicated that employees resign their organization for a variety of reasons; these can be classified into the following parts.

2.3.1 Demographic Factors:

Studies focus on the demographic factors to see whether turnover is associated to age, marital status, gender, number of children, education, experience and employment tenure (Spector,

2008). Empirical studies have demonstrated that turnover is associated in particular situations with demographic and biographical characteristics of workers. Several studies have reported negative relationship between turnover and three demographic factors, age, tenure, and income level (Arnold & Feldman, 1982); (Cotton & Tuttle, 1986); (Mobley et al., 1979); (Price & Mueller, 1986).

According to Kondalkar (2007), the most significant factors among demographic factors to affect turnover are listed as follows.

- **i. Age**: Age is one of the important factors to generate productivity. Job performance and turnover declines with increasing age and therefore less job opportunities, hence there are less of resignations. Older people generally get higher wages, longer paid vacation and they stick on to the job because of pensionary benefits.
- **ii. Marital Status:** Marriage imposes increased responsibilities on an individual; therefore an individual desires to have a steady job. Married employees have fewer absences, lesser turnover and higher job satisfaction. Even if there is no clear cut information about the impact of marital status on productivity, absenteeism, turnover and job satisfaction, since marriage increases the family commitment, it is expected for an individual to not change organizations.
- **iii. Tenure:** Tenure is referred to as experience. There is a positive relationship between seniority and job performance and negative relationship between seniority and absenteeism. Employee turnover is negatively related to seniority. Research indicates that experience and satisfaction are positively related. Seniority and experience is a better indicator of job satisfaction than the age of a person.

Level of education, on the other hand, is found to be positively associated with turnover suggesting that the more educated employees quit more often (Cotton & Tuttle, 1986).

2.3.2 Personal Factors

Personal factors such as health problem, family related issues, children education and social status contributes in turnover intentions. There is another important variable "Job-Hoping" which also contributes in turnover intentions. When there is a labor shortage, employees have plenty of jobs available. Consequently, they can afford to switch jobs for a few additional money. Unrealistic expectation of employee's is also an important personal factor which contributes in turnover. Many people keep unrealistic expectations from organization when they join. When these unrealistic expectations are not realized, the worker becomes disappointed/dissatisfied and

quit. Masahudu (2008) cited in Ali Shah et al. (2010) has identified "employers' geographic location" as important variable that may determine turnover. The closeness of employees to their families and significant others may be a reason to look elsewhere for opportunities or stay with their current employers.

2.3.3 Pull Factors

Pull factors are those reasons that attract the employee to a new place of work. In some researches pull factors are named as uncontrolled factors because it is out of the control of organizations. So in this category we would have, a better paying job, a career advancement opportunity that they wouldn't have got in the short term had they stayed with their present employer, new challenge and interesting work, good location of company, life-work balance and more benefits and good boss and so on.

2.3.4 Push Factor

Push factors are aspects that push the employee towards the exit door. In the literature it is also called controlled factors because these factors are internal and can be controlled by organizations. According to Loquercio (2006) (cited in Ali Shah et al, 2010), it is relatively rare for people to leave jobs in which they are happy, even when offered higher pay elsewhere. However, sometimes employees are 'pushed' due to dissatisfaction in their present jobs to seek alternative employment. On the basis of available literature, push factor can be classified into:

2.3.5 Organizational Factors

There are many factors which are attached with an organization and work as push factors for employees to quit. Among them which are derived from various studies are: salary, benefits and facilities; size of organization (the number of staff in the organization); location of the organization (small or big city); nature and kind of organization; stability of organization; communication system in organization; management practice and polices; employees' empowerment and organizational justice. According to Stredwick (2005) organizational justice means fairness in the workplace. There are two forms of organizational justice: distributive justice, who describes the fairness of the distribution of pay in line with employee's worth and output; and procedural justice, describes the fairness of the procedures used to determine those outcomes.

2.3.6 Attitude Factors

In the literature, attitude is another kind of push factor which is mostly attached with employee behavior. Attitude factors are further classified into job stress and job satisfaction. Job stress

includes variables such as role ambiguity, role conflict, work-overload and work-family conflicts that creates stress on the individual employee which pushes him to leave.

Job satisfaction on the other hand, can be defined as an individuals' total feeling about their job and the attitudes they have towards various aspects or facets of their job, as well as an attitude and perception that the individual has towards the organization (Spector, 1997).

Job satisfaction has been the most frequently investigated variable affecting job performance and organizational effectiveness in human resources and organizational behavior and management.

(Spector, 1997) states that job satisfaction influences people's attitude towards their jobs and various aspects of their jobs. The consequences of job satisfaction include better performance and a reduction in withdrawal and counter-productive behaviors. Since job satisfaction involves employees' emotions, it influences an organization's well-being with regard to job productivity, employee turnover, absenteeism and life satisfaction.

There are two types of job satisfaction which are based on the level of employees' feelings regarding their jobs. The first, and most studied, is global job satisfaction, which refers to employees' overall feelings about their jobs. The second is job facet satisfaction, which refers to feelings about specific job aspects, such as salary, benefits, and the quality of relationships with one's co-workers. According to Kerber and Campbell (1987) measurements of job facet satisfaction will be helpful in identifying which specific aspects of a job require improvements. The results will aid organizations in improving overall job satisfaction or in explaining organizational issues such as high turnover.

In previous several researches it is found that job satisfaction and turnover are inversely related. Turnover is high when total job satisfaction is low, while when satisfaction remains at a high level, employee turnover is low. Nevertheless, it must be stressed that this relationship is not necessarily observed in all circumstances. Some people remain in their jobs although they do not feel satisfied. This is mainly observed in situations where unemployment rate is high, few job are available and, consequently, the cost of quitting for the individual is high (Peerbhai, 2006) (cited in Chatzoglou, 2011).

2.4 Effects of Employees Turnover

Turnover leads to various costs for the organization to bear. Some of them are:-Severance pay, advertisings and recruiting to replace the departed employee, orientation and training for a new employee, lost production time during the process of recruitment and replacement, and lower

productivity while the new employee learns the job. Other costs to bear are effluent from the fact the resigned employees may take with them their valuable tacit knowledge and expertise gained during the work in an organization (Bluedorn, 2001). Quite often leaving employees also can take together with them clients from the former organization (Mitchell et al., 2001).

2.5 Methods of Identifying Reasons for Staff Turnover

Since turnover involves the most important resource of an organization, it needs to be examined and monitored. Organizations need to know who is leaving, why they are leaving, and whether any effort on their part can slow turnover. Several different methods are available to researchers seeking to investigate why employees choose to leave. Here we consider four contrasting approaches: exit interviews, survey of ex-employees, attitude surveys, and quantitative approaches (Taylor, Stephen, 1998).

2.5.1 Interview

Undertaking formal interviews with employees before they leave the organization is a common method used to develop an understanding of their motivation for resigning. The most straightforward approach is to take the resigned through a questionnaire of direct question concerning his/her satisfaction with pay, supervisor, development opportunities, relationships with colleagues and job content. There are, however, a number of problems with such approaches that can serve to reduce their effectiveness. First, there is the tendency of employees to develop a far more optimistic outlook after they have secured a new job and resigned. Their original reasons for seeking alternative employment offer get forgotten as they move toward their last day.

Such feelings are compounded if counter-offers are made to encourage them to stay and may disappear completely in the last days as cards are signed, leaving presents bought, affectionate speeches given and farewell parties held. This is often not, therefore, the best time to ask them for an honest and well-balanced assessment of their reasons for quitting (Taylor, Stephen, 1998). A further problem arises when supervisors or department heads undertake exit interviews, because leavers will often hesitate to implying any criticism of them-particularly if they believe that they will require positive references in the future. The reason given for leaving may thus obscure the whole truth or may even be entirely false, (ACAS, 2005). It can thus be argued that exit interviews, if used at all, should be undertaken very soon after the resignation has been

confirmed, and that they should be carried out by an individual who will not have any role in writing future job references (Taylor, Stephen, 1998).

2.5.2 Surveys of Ex-employees

Another way of collecting information about the reason for staff resignation is to contact former employees some months after they left the organization and ask them for a considered view of their reasons for resignation. While the use of this method is relatively rare, as cited by Stephen Taylor, there have been a number of cases covered recently in the personnel journals that indicate some large organizations are experimenting with it. Openness is further encouraged if the surveys are carried out by independent bodies and are clearly labeled "private and confidential".

2.5.3 Attitude Surveys

A third approach is to seek the views of employees before they leave and so provide a basis for the development of policies and practices that will deter them from so doing. These too are truly effective only if confidential-so as to maximize the chance of employees' stating honestly how they feel about their jobs, their perceived opportunities, their bosses, colleagues and the organization as a whole. Questions can also be asked about their current intentions as regards the future and about their perception of alternative career paths open to them. Such approaches enable employers to anticipate in which areas future turnover is most likely to occur, and to gain an insight into the main causes, (Taylor, Stephen, 1998).

2.5.4 Quantitative Approaches

An alternative method to the use of surveys is to make use of the employee records to compare the data or characteristics of those who leave with those who stay. Although quantitative approaches are unlikely in themselves to give a particularly clear picture of reasons for turnover, they may reveal some interesting general trends and can usefully supplement information gathered using the three other methods outlined above.

2.6 Reducing Labour Turnover

Based on the review of literature the list of practices used to reduce labour turnover that capture the main types of interventions are:

2.6.1 Compensation and Benefits

Compensation Levels:-Competitive compensation packages can signal strong commitment on the part of the organization, and can therefore build a strong reciprocal commitment on the part of workers. However, to the extent that it contributes to retention, competitive compensation is

also likely to affect both desirable and undesirable turnover: it will help to retain workers, irrespective of the quality of their contribution to the company (Lochhead C. & Stephens A.,2004).

While there is general agreement about the importance of competitive compensation for employee retention, there is also a growing consensus that competitive, or even generous compensation will not single-handedly guarantee that a company will be able to keep its most valuable employees (Lochhead C. & Stephens A., 2004).

Building "affective commitment" Meyer et al (2003) involves much more than paying well, and that retention based on the principle of "compensation-based commitment" is sensitive to changes in compensation within the organization. Employers that base their retention on compensation-based commitment will always be vulnerable to the possibility that their competitors will be able to offer better wages and thus lure away their employees.

Companies that wish to develop a successful retention plan that includes compensation and benefits must always understand their own unique characteristics and circumstances (Lochhead C. & Stephens A., 2004).

According to Branham (2000) compensation can also drive commitment-oriented behaviour by: Sending employees a strong message about what results are valued, Recognize and reinforce important contributions so that employees feel valued, Provide a sense of "emotional ownership" and increased commitment that comes from giving employees "a piece of the action," Maintain or cut fixed payroll costs and allocate variable pay to the employees they most want to attract and keep.

Benefits: - Benefits can demonstrate to employees that an organization is supportive and fair, and there is evidence to suggest that stable benefits are at the top of the list of reasons why employees choose to stay with their employer or to join the organization in the first place (Tompkins and Beech, 2002).

Employee benefits are constantly evolving as the workforce itself evolves, and as people identify new priorities as being important. It is important to note that the relative importance of benefits will vary according to the specific needs of each individual (Lochhead C. & Stephens A. 2004). The importance of a benefit plan to an employee with dependents may assume far greater importance particularly in an environment where benefits costs are increasing rapidly (Tompkins and Beech, 2002). Organizations with an older workforce demographic might see their retention

efforts best served through benefit plans that cover health-related expenses, care for elderly parents, etc. Companies with a significantly younger demographic may find education subsidies or tuition rebate to be more effective retention drivers.

Performance-based Compensation: In general, the key purpose behind performance-based compensation is twofold. It is (i) a means of modifying individuals' behaviour within an establishment in order to better align their activity with particular business objectives, and (ii) a way of influencing the development of particular types of organizational culture e.g., team based pay can lead to a stronger team-based culture, skill-based pay, or Retention-based Compensation can contribute to a culture of stability and mutual commitment, etc.; (Meyer, 2003).

The major advantage of performance-based pay system is that it can promote a culture of flexibility. Individuals can perform multiple tasks, including filling jobs in the wake of turnover. Skill-based pay can also contribute to lower turnover since individuals will be paid more for their knowledge and will be unlikely to find similarly attractive pay in firms using more traditional job-based pay systems.

2.6.2. Recognition and Rewards

Rotundo (2002) argues that reward systems ought to be a significant sphere of innovation for employers. The increasing diversity of the workforce, she says, suggests the need for more creative approaches to tailoring the right rewards to the right people. It would, however, be impossible to list all of the various types of recognition and rewards that companies actually give their workers. It is clear that recognition and rewards—as part of a more comprehensive effort at keeping workers or adopting good workplace practices—can contribute to increased retention.

2.6.3 Training, Professional Development and Career Planning

Training and development are so enthusiastically embraced as key factors to good retention with no doubt due to the fact that well-developed training programs are becoming ever more essential to the ongoing survival of most modern organizations, whether or not retention is an important issue to that organization. To the extent that operational paradigms such as "The Learning Organization" or the "Knowledge-Based Organization" continue to take hold in the contemporary business world, training is only likely to become more important (Lochhead C. & Stephens A., 2004). In any event, retention reflects a desire to keep one's valued people; but it is just as much about keeping and managing the skills that a company needs to meet its goals. The provision of training is a way of developing those skills in the first place. The fact that providing

it also turns out to be a benefit that is highly valued by those who receive it makes for a very powerful approach to doing business (Lochhead C. & Stephens A., 2004).

Because training and professional development are so fundamental to the operation of an organization, it goes against intuition to suggest that training and development are to be thought of primarily as "retention" tools. The evidence seems to confirm that the link between training and retention is even stronger for more highly skilled workers (Paré et al, 2000).

Meyer et al. (2003) suggest that employee learning - which encompasses training and development but is also related to socialization within the workplace - contributes to retention by: Building employee commitment through a show of support, Providing employees with the means to deal with stress related to job demands and change, Serving as an incentive to stay, and creating a culture of caring.

Thus, training and professional development are seen as ways of building employee commitment in that they allow employees to "see a future" where they work, and provide them with the support necessary to face the on-going challenges related to their work.

Many employers will voice the familiar concern that there is a risk that once trained workers may be tempted to leave the company for other opportunities. This is no doubt a valid concern amongst many employers, particularly those in large industrial centers. Lochhead C. & Stephens A. (2004) As such, companies are only willing to provide training at the minimum level or to provide more extensive training on proprietary equipment and processes. The picture may be somewhat different for more highly-skilled segments of the workforce, where employers appear to be more willing to make greater investments.

There is a good fit between training and a number of other retention-related practices, such as career development and planning, skill-based pay, and others. Taken together such practices can usefully complement one another. Training can be a particularly strong retention tool when it is combined with measures designed to allow people to develop and progress within a company (Meyer et al, 2003). It sends a very strong signal to employees that they are important to the company, and that the company is willing to make real investments in keeping them there. In addition, the advantage to training-related incentives is that they can allow companies to focus particularly on employees who demonstrate a strong potential for growth and contribution to the company.

Such arrangements end up being to the advantage of both parties. The company is able to secure the skills it needs while at the same gaining some reassurance that its skills development investments are also creating the conditions that will keep those skills in the company. The employee, meanwhile, gains the opportunity to securely move towards better pay and more challenging responsibilities.

Training is always unique to the circumstances of each company and that there is no formula that dictates how much and what kind of training is given. It does seem plausible that training, when combined with well-communicated plans for advancement and ongoing professional development within the company, can help organizations to keep their valued employees (Lochhead C. & Stephens A., 2004).

2.6.4 Recruitment and Orientation

How organizations recruit and how they provide orientation in the first days on the job can be of crucial importance to keeping workers over the longer term. Failure to effectively recruit and orient employees may impose significant separation and replacement costs down the road (Lochhead C. & Stephens A., 2004). Smith (2001) attributes 60% of undesirable turnover to bad hiring decisions on the part of the employer. "Bad hiring decisions" may cover a number of considerations, including overly hasty selection processes that fail to ensure that the job candidate really has the adequate skills and qualifications to do the job for which she or he is hired. Research confirmed that good employee retention is in part a result of a good "fit" between a company's workplace culture — its way of doing business and the qualities that it espouses as valuable — and the interests, character, and motivations of the individuals that exist within it. In terms of recruitment, companies should therefore put an emphasis on not only evaluating formal qualifications, job-relevant technical ability, etc., but also more general types of qualifications and dispositions on the part of the recruit. If work in the company involves being part of a highly cohesive team, the company may want to recruit individuals who are interested in and capable of working in such an environment. (Lochhead C. & Stephens A., 2004)

It is important for the individual job candidate to have a fair chance at deciding whether the company is a good fit for him or her. Meyer et al. (2003) have emphasized the importance of providing "realistic job previews" to potential employees. These previews provide potential new hires with more than just a cursory glance at the organization's operations, providing the

candidate with enough information to make a decision about whether it is the right workplace for her or him. The organization must have at its disposal the right tools to enable it to make reasonably accurate evaluations about the type of people it might hire; and there exist a great number of evaluation tools at the employer's disposal.

Including one's employees in helping to evaluate candidates may also be particularly effective with respect to retention in workplaces where team-based work is the norm (Meyer, 2003).

Allowing employees to have a say in who they will be working with, and asking for their assistance in evaluating whether that person will be a good fit, may prove helpful in ensuring that the candidate not only has the requisite experience but will also be an effective member of the work team.

2.6.5. Healthy Workplace and Well-being Programs

There is a growing body of evidence that workplace safety, health and wellness initiatives can make a fundamental contribution to business performance as well as the improved health and well-being of individual employees. Healthy workplace practices take on a variety of forms, including those directed at the physical work environment (safety, ergonomics, etc.); health practices (supporting healthy lifestyles, fitness, diet, etc.); and social environment and personal resources (organizational culture, a sense of control over one's work, work-family balance, etc.). Thus, to the extent that emotional and mental wellness is important to good retention and other indicators of business performance, organizations might be well-advised to turn their attention to factors in the workplace that cause stress in the workplace (Lochhead C. & Stephens A., 2004).

2.6.6. Work-Life Balance

Work-life balance programs cover a variety of interventions, and include such practices as dependent care leave, childcare subsidies, eldercare programs, counseling and referral, and flexible working hours (Withers, 2001). As the list suggests, the concept of "work-life balance" recognizes that employees have important family and extraprofessional obligations that compete with their professional commitments; benefits that may be grouped under this concept therefore allow people to strike a more meaningful and potentially less stressful balance between obligations at the workplace and obligations at home. Duxbury & Higgins (2001) note that the 1990s saw an intensification of work-life conflicts for many workers. Jobs have become more stressful and less satisfying, and employees generally exhibit less commitment to their employers and higher absence rates from work. They add that high levels of "role overload" and "work to

family interference" play a significant role in frustrating recruitment and retention in organizations. Many employers have begun to respond to such demands by implementing "worklife balance" initiatives in their workplace, and have done so out of a conviction that providing such benefits can substantially enhance productivity, revenues, and employee retention and commitment (Withers, 2001)

The key to success in this area appears to be, first and foremost, a good channel of communications and a workplace culture in which employees feel comfortable in asking for time off to deal with pressing family matters, and employers are willing to recognize that granting time off in such a manner ultimately may contribute to greater employee commitment and productivity in the long run. Duxbury and Higgins (2001) argue that employers can help to create more supportive work environments by:

(i) Working with employees to identify and implement the types of support they need, and better inform them about policies that may currently be available to them, and (ii) Encouraging employees to use the supports that are readily available and ensure that employees who could make use of such assistance do not feel that their career prospects would be jeopardized by doing so (Duxbury and Higgins, 2001).

2.6.7 Job Design and Work Teams

With respect to retention, "intrinsic" rewards are just as important as material rewards. That is workers value their jobs not only when they are well compensated, but often because doing the job is in itself a rewarding experience — in other words, it is fulfilling, challenging, interesting and stimulating. Practices such as autonomous or semi-autonomous work teams, 'self scheduling, and job rotation can not only improve retention but have also been shown to improve a number of other important indicators such as productivity, accidents and injuries and product quality (Lochhead C. & Stephens A., 2004).

Based on the theoretical and empirical literature the factors that make employees to be satisfied includes interesting work, fair and competitive pay, good promotional opportunities, supportive and considerate supervisor, friendly co-workers, justifiable policies and procedures and satisfaction in the information flow within the organization. As per Brief (1998), (cited in Armstrong, 2007) "if the pleasures associated with one's job outweigh the pains, there is some level of job satisfaction".

variables and employee turnover.

Therefore, this research would indicate a connection between these previously identified

CHAPTER THREE

3.1. RESEARCH DESIGN AND METHODOLOGY

This chapter explains the way in which the research will be carried out. How the research problem was explored, with specific reference made to how the participants were selected and the procedure followed to gather the data. In addition, the measuring instruments to gather the data and the statistics technique utilized are discussed.

3.1.1. Research Design

The research design helps the researcher to obtain relevant data to fulfill the objectives of the study (Churchill and Iacobucci, 2002). In order to fulfill the research objectives, it is important that research is designed in approach that data collected in both accurate and relevant. Kinner et al (1996) suggested that a research design is the basic plan that guides the data collection and analysis. It is the structure that specifies the kind of information to be collected, the sources of data and the data collection procedure. An examination of research literatures suggests that research design can be classified according to the research objectives or the type of research being employed. This study will employ a descriptive design in collecting primary data.

3.2. Population and Sampling Techniques

3.2.1. Population

Target population is the specific, complete groups that are relevant to the research. The aim of this research is to identify the factors influencing employee turnover intention among NIB employees in Addis Ababa. NIB has 104 branch offices including the Headquarter. Of the 104 branches, 62 branches are located in Addis Ababa and the rest in the different regions of the country. The study area is the head office and 21 purposefully selected branches located in Addis Ababa only. The justification for purposefully select these 21 branches of the organization is because these branches have several employees compared to other branches. Moreover the selection considers the representation of both customer service and back office category of the work and non-professional and professional workers, the target population for this research would be 210 staffs working in these 21 branches. Minimum of 10 staffs per each branch would be considered 2 managerial staffs from each branch total of 42 respondents 6 clerical but not managerial staffs per branch total of 126 respondents and 2 non-clerical staffs in total of 42 respondents would be involved.

3.2.2. Sampling Techniques

Stratified sampling technique is used to select samples from the existing employees of the organization; this is because, the study focuses on different job grades of respondents and each group of the respondents is required to have its own representative from the total sample size. From each stratum respondents are selected through random sampling method, from management body sample respondents were selected purposely (Purposive sampling method) and the representative of ex-employees selected through convenience sampling method.

In NIB there are 21 job grades. The strata of the existing employees, therefore, are equally distributed among these 21 job grades from the total population.

In determining the sample size from total population (Naresh, 2007) was used and the researcher believed and selected the sixth stage because the total population is about 2582.

3.2.3. Sample Determination

Table 1 Sample Determination

Population	Sample Size
1 opulution	~

	Low	Medium	High
51-90	5	13	20
91-150	8	20	32
151-280	13	32	50
281-500	20	50	80
501-1200	32	80	125
1201-3200	50	125	200
3201-10,000	80	200	315
10,001-35,000	125	315	500
35001-150,000	200	500	800

Source; Naresh, M. (2007). Marketing Research: an applied approach

3.3. Research data sources.

There are two basic types of data further on which research can be originated, primary and secondary data.

3.3.1. Primary data research

The research will use primary data collection method; first-hand data will be collected through survey questionnaires. The data acquiring process will have few stages; namely, identification of areas for questionnaire distribution, distribution of questionnaires, reminding respondents of questionnaire deadline and gathering of questionnaire.

3.3.2. Secondary data research

Secondary data research concerns the examination and evaluation of existing data collected by previous research projects. It can be available to researchers on two types internal and external data. Kinner et al (1996) describes internal data is that exists within an organization itself and comes from sources such as organization figures, reports and past research.

External secondary data relays to sources such as reports and statistics from the government agencies, journal and textbooks. Proctor (2000) suggested that also internet appeared as a data

source assists the collection of secondary data is available through online reports, journals and magazines because of several websites that have been created by government agencies and companies. In this method, this study will use only reliable information sources in order to present applicable, appropriate research sources.

3.3.3. Research questionnaires

Sarantakos, (1998), explains questionnaire as a method for collecting and analyzing social data via highly structures and often very detailed interviews or questionnaires in order to obtain information from large numbers of respondents presumed to be representative of a specific population."

Saunders, (2007) Questionnaires can be used for descriptive and explanatory research. Descriptive research such as that undertaken using attitude and opinion questionnaires and questionnaires of organizational practices will enable researcher to identify and describe the variability in different phenomena.

3.3.4. Procedures of Data Collection

Questionnaire survey is the most common method and it will be adopted in this research. The questionnaire consists of nine sections: Demographic factors such as age, educational level, number of years with current organization, job grade in NIB, marital status, gender and category of work will be in (Section I); main questioner part will be coded as 1= strongly disagree, 2=disagree,3=neutral,4= agree and 5= strongly disagree and these dimensions are: pay, promotion (Section II), supervision (Section III), benefit (Section IV), operating procedure (Section V), Coworkers (Section VI), nature of work (Section VII), communication (Section VIII) and turnover intention (section IX).

Table Two

Variables	Number of Items	Source of the Instruments
Pay	4	(Smith, 1976)
Promotion Opportunity	4	(Tate et al. 1997)
Supervision	7	(Smith 1976)
Benefit and Rewards	8	(Folger&Konovsky, 1989)

Operating Procedure	4	(Magner et al., 1994)
Coworkers	4	(Weiss et al., 1967)
Nature of work	7	Weiss et al., 1967
Communication	4	(Smith, 1976)
Turnover Intention		Cummann et al, 1979

3.3.5. Analysis of data

The researcher used deductive based analytical procedures to analyses data where the researcher used the Causal mode of turnover as the framework for analyzing data. The deductive requires a greater structure and set of procedures to follow where data categories and codes to analyze data are derive from theory and analytical framework, Saunders, 2007).

3.3.6. Descriptive Analysis

Descriptive statistics describes data through tables, diagrams and graphs. In this research, the descriptive analysis used to analyze the respondent's demographic information through tables in addition to general research issues.

3.3.7. Research limitations

The specific problems and limitations that had been experienced in undertaking the research identified after conducting the research. Nevertheless, this chapter set out some identified limitations of the research design and general possible limitations predicted in carrying out the study. The predictable research limitations include the following.

.

3.4. Time restraints.

This is major concern when identifying the case study as a researcher has to conduct research at the head quarter and branches.

3.5. Access to key information and information holders

The case study strategy in particular requires an in depth understanding that only practical if key practitioners are prepared to cooperate by providing access to documents and taking part in interviews, even where people are willing to participate it can be difficult co-coordinating meeting.

3.6. Study Model

The study used regression model to present ad combine the results of the collected data on the suitable way of data analysis. The main model considered for the research has been formulared like.

$$TI = \alpha TI + \beta 1 \text{ (SP)} + \beta 2 \text{ (BR)} + \beta 3 \text{ (CW)} + \beta 4 \text{ (NW)} + \beta 5 \text{ (PR)} + \beta 6 \text{ (CM)} + \epsilon$$

Where

TI = Turnover Intension

SP = Supervision

BR = Benefits and Rewards

CW = Co-Workers

NW = Nature of Work

PR = Promotion

CM = Communication

E = Standard error

 αTI = The regression constant

 $\beta 1,\,\beta 2,.....$ βi the partial regression coefficient for the independent variable

CHAPTER FOUR

4. Data Presentation and analysis

This chapter discusses the analysis of the data collected and the results of the regression analysis.

4.1. Survey results

All questioners were distributed to those selected banks branches, as a result 125 questioners returned successfully.

Sample size and returned questioners.

Table 4.1.1 sample size

Sample size	150
Successfully returned questioners	125
Response rate	83.33%

Out of one hundred fifty total distributed questioners, one hundred twenty five questioners collected successfully and we can say that 83.33% of the total target population has been involved in giving the realistic view of various thoughts and feelings on the selected study area respective of their organization.

Table 4.1.2 Respondent's educational background

Education level	
Degree and above	88
Diploma	20
Certificate	17
Total	100%

Out of those one hundred twenty five respondents, eighty eight respondents were first degree and above holders, twenty respondents were diploma holders and seventeen respondents were certificate holders.

Table 4.1.3. Length of stay in current bank of the respondents in years

Length of stay	Sum	Rate
Below two years	24	19.2%
2-5 years	49	39.2%
6-10 years	39	31.2%
Greater than ten years	13	10.4%
Total	125	100%

In respect of the length of stay in the bank, twenty five respondents were in below two service years, forty nine respondents have between two and five service years, thirty nine respondents were in six and ten service years in the bank and only thirteen respondents were in more than ten service years.

4.2. Descriptive analysis in terms of frequency per each variable Statistics

For all Variables

N	Valid	125
	Missing	0

Frequency Table

Table 4.2.1. Fairpay

	Frequenc y	Percent	Valid Percent	Cumulative Percent
strongly disagree	4	3.2	3.2	3.2
Validslightly disagree	49	39.2	39.2	42.4
neutral	41	32.8	32.8	75.2
slightly agree	29	23.2	23.2	98.4
strongly agree	2	1.6	1.6	100.0
Total	125	100.0	100.0	

Regarding the fair pay assumption of employees on their contribution to the bank, 3.2% of the respondents strongly disagreed, 39% of the respondents slightly disagreed, and 32.8% of the respondents were neutral and only 24% of the respondents agreed on their banks fairness on salary payment. As this data can explain

There is less remuneration or fair pay system in the bank; employees are dissatisfied with current pay structure of NIB.

Table 4.2.2.Appreciation

		Frequenc	Percent	Valid	Cumulative
		y		Percent	Percent
Valid	strongly disagree	7	5.6	5.6	5.6
	slightly disagree	48	38.4	38.4	44.0
	neutral	51	40.8	40.8	84.8
	slightly agree	17	13.6	13.6	98.4
	strongly agree	2	1.6	1.6	100.0
	Total	125	100.0	100.0	

On the subject of the Appreciation for good work, 5.6% of the respondents strongly disagreed, 38.4% of the respondents slightly disagreed, 40.8% of the respondents were neutral, 13.6% of the respondents slightly agree and only 1.6% of the respondents strongly agreed.

There is lack of appreciation for good work done by employees, which implies that employees are doing their job without expecting some recognition and appreciations from their supervisors.

Table 4.2.3.Promotion

		Frequenc	Percent	Valid Percent	Cumulative Percent
	atmam alar	y		1 CICCIII	1 CICCIII
Valid	strongly disagree	6	4.8	4.8	4.8
	slightly disagree	46	36.8	36.8	41.6
	neutral	48	38.4	38.4	80.0
	slightly agree	24	19.2	19.2	99.2
	strongly agree	1	.8	.8	100.0
	Total	125	100.0	100.0	

For promotion 4.8% of the respondents strongly disagreed, 36.8% of the respondents slightly disagreed, 38.4% of the respondents were neutral, 19.2% of the respondents slightly agree and only .8% of the respondents strongly agreed.

Most Employees are not benefited from timely promotion within the bank even if there experience and stay in the bank deserves.

Table 4.2.4.Competent supervisor

	Frequenc	Percent	Valid	Cumulative
	У		Percent	Percent
strongly disagree	7	5.6	5.6	5.6
Validslightly disagree	62	49.6	49.6	55.2
neutral	38	30.4	30.4	85.6
slightly agree	18	14.4	14.4	100.0
Total	125	100.0	100.0	

Regarding competent supervisor, 5.6% of the respondents strongly disagreed, 49.6% of the respondents slightly disagreed, 30.4% of the respondents were neutral and 14.4% of the respondents slightly agreed.

The bank has lack of competent supervisors in various work units as per the view of the respondents.

Table 4.2.5.Littleinterestforsubordinates

		Frequenc	Percent	Valid	Cumulative
		У		Percent	Percent
Valid	strongly disagree	2	1.6	1.6	1.6
	slightly disagree	31	24.8	24.8	26.4
	neutral	54	43.2	43.2	69.6
	slightly agree	37	29.6	29.6	99.2
	strongly agree	1	.8	.8	100.0
	Total	125	100.0	100.0	

About little interest for subordinates, 1.6% of the respondents strongly disagreed, 24.8% of the respondents slightly disagreed, 43.2% of the respondents were neutral, 29.6% of the respondents slightly agree and only 0.8% of the respondents strongly agreed.

Employees have good feelings for their subordinates within the bank which can strengthen the maintenance of good relationship within the bank.

Table 4.2.6. Easy communication with supervisor

	Frequenc y	Percent	Valid Percent	Cumulative Percent
strongly disagree	4	3.2	3.2	3.2
Validslightly disagree	56	44.8	44.8	48.0
neutral	27	21.6	21.6	69.6

slightly agree	37	29.6	29.6	99.2
strongly agree	1	.8	.8	100.0
Total	125	100.0	100.0	

Regarding easy communication with supervisors, 3.2% of the respondents strongly disagreed, 44.8% of the respondents slightly disagreed, 21.6% of the respondents were neutral, 29.6% of the respondents slightly agree and only 0.8% of the respondents strongly agreed.

Most employees are suffering from challenges for easy communication with their supervisors.

Table 4.2.7.Rulesandprocedures

	Frequen cy	Percent	Valid Percent	Cumulative Percent
strongly disagree	6	4.8	4.8	4.8
slightly disagree	40	32.0	32.0	36.8
Validneutral	52	41.6	41.6	78.4
slightly agree	26	20.8	20.8	99.2
strongly agree	1	.8	.8	100.0
Total	125	100.0	100.0	

On the topic of rules and procedures application and distribution, 4.8% of the respondents strongly disagreed, 32% of the respondents slightly disagreed, 41.6% of the respondents were neutral, 20.8% of the respondents slightly agree and only 0.08% of the respondents strongly agreed.

Various policies and procedures of the bank are not distributed in a wide range of distribution, most of employees fail to response for the effectiveness of the bank's operational policies and procedures.

Table 4.2.8.Toomuchwork

Frequenc	Percent	Valid	Cumulative
У		Percent	Percent

Valid	strongly	4	3.2	3.2	3.2
vanu	disagree	4	3.2	3.2	3.2
	slightly disagree	42	33.6	33.6	36.8
	neutral	42	33.6	33.6	70.4
	slightly agree	37	29.6	29.6	100.0
	Total	125	100.0	100.0	

On the topic of too much work, 3.2% of the respondents strongly disagreed, 33.6% of the respondents slightly disagreed, 33.6% of the respondents were neutral and 29.6% of the respondents slightly agreed.

Work volume varies from post to post of several work units of the bank, there are employees who are so loaded with too much work and vice versa.

Table 4.2.9.Incompetenceofpeople

	Frequenc y	Percent	Valid Percent	Cumulative Percent
strongly disagree	5	4.0	4.0	4.0
Validslightly disagree	47	37.6	37.6	41.6
neutral	45	36.0	36.0	77.6
slightly agree	28	22.4	22.4	100.0
Total	125	100.0	100.0	

When incompetence of people is evaluated, 4% of the respondents strongly disagreed, 37.6% of the respondents slightly disagreed, 36% of the respondents were neutral, 22% of the respondents slightly agreed.

There are competent staffs with in the bank.

Table 4.2.10. Fighting atwork

Frequenc	Percent	Valid	Cumulative
у		Percent	Percent

strongly disagree	5	4.0	4.0	4.0
Validslightly disagree	51	40.8	40.8	44.8
neutral	39	31.2	31.2	76.0
slightly agree	29	23.2	23.2	99.2
strongly agree	1	.8	.8	100.0
Total	125	100.0	100.0	

Regarding fighting at work, 4% of the respondents strongly disagreed, 40.8% of the respondents slightly disagreed, 31.2% of the respondents were neutral, 23.2% of the respondents slightly agreed and only 0.08% of the respondent strongly agreed.

There is little fight at work between some employees of the banks, the bank has not that much problem from such a like problems.

Table 4.2.11.Clearresponsibilities

	Frequen cy	Percent	Valid Percent	Cumulative Percent
strongly disagree	9	7.2	7.2	7.2
Validslightly disagree	56	44.8	44.8	52.0
neutral	32	25.6	25.6	77.6
slightly agree	26	20.8	20.8	98.4
strongly agree	2	1.6	1.6	100.0
Total	125	100.0	100.0	

Regarding clear responsibilities, 7.2% of the respondents strongly disagreed, 44.8% of the respondents slightly disagreed, 25.6% of the respondents were neutral, 20.8% of the respondents slightly agreed and only 1.6% of the respondent strongly agreed.

Most of the responsibilities and tasks given for employees are clear and easily understandable, which employees can do them without spending too long time and resources.

Table 4.2.12. Enough time

1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1					
	Frequenc	Percent	Valid	Cumulative	
	y		Percent	Percent	

strongly disagree	5	4.0	4.0	4.0
Validslightly disagree	54	43.2	43.2	47.2
neutral	35	28.0	28.0	75.2
slightly agree	30	24.0	24.0	99.2
strongly agree	1	.8	.8	100.0
Total	125	100.0	100.0	

On the subject of enough time given for work, 4% of the respondents strongly disagreed, 43.2% of the respondents slightly disagreed, 28% of the respondents were neutral, 24% of the respondents slightly agreed and only 0.8% of the respondent strongly agreed.

The time given for various tasks in numerous work units is not enough to complete as per available resources and employees capability to do as it is expected from them.

Table 4.2.13. Frustrationatjob

	Frequenc y	Percent	Valid Percent	Cumulative Percent
strongly disagree	3	2.4	2.4	2.4
Validslightly disagree	49	39.2	39.2	41.6
neutral	46	36.8	36.8	78.4
slightly agree	26	20.8	20.8	99.2
strongly agree	1	.8	.8	100.0
Total	125	100.0	100.0	

Regarding frustration at job, 2.4% of the respondents strongly disagreed, 39.2% of the respondents slightly disagreed, 36.8% of the respondents were neutral, 20.8% of the respondents slightly agreed and only 0.08% of the respondent strongly agreed.

Frustration at job by employees raised from various reasons is widely disseminated within the bank.

Most wok units of the bank needs plenty of time to be done as a result employees are unable to spend their slack time with their family and relatives.

Table 4.2.14.Clear organizational goals

		Frequenc	Percent	Valid	Cumulative
		y		Percent	Percent
Valid	strongly disagree	4	3.2	3.2	3.2
	slightly disagree	44	35.2	35.2	38.4
	neutral	55	44.0	44.0	82.4
	slightly agree	22	17.6	17.6	100.0
	Total	125	100.0	100.0	

When the respondents are asked for their organization goal clearness, 3.2% of the respondents strongly disagreed, 35.2% of the respondents slightly disagreed, 44% of the respondents were neutral, 17.6% of the respondents slightly agreed.

Most of the bank's goals are not clear for greatest number of employees in the bank.

Table 4.2.15.Lookingfornewjob

		Frequenc	Percent	Valid	Cumulative
		y		Percent	Percent
Valid	strongly disagree	3	2.4	2.4	2.4
	slightly disagree	41	32.8	32.8	35.2
	neutral	40	32.0	32.0	67.2
	slightly agree	41	32.8	32.8	100.0
	Total	125	100.0	100.0	

When respondents are asked for their feelings on looking for new job, 2.4% of the respondents strongly disagreed, 32.8% of the respondents slightly disagreed, 32% of the respondents were neutral, 32.8% of the respondents slightly agreed.

Most of workers of the bank are looking for new job.

4.3. Descriptive Statistics

In the descriptive statistics important observations related to the dependent and independent variables has been made. The dependent variable is employee turnover intention and independent variables are promotion, supervision, benefits and rewards, co-workers, nature of work and poor communication.

The descriptive statistics for selected variables is shown in Table 4.3.1 below.

Table 4.3.1.Descriptive Statistics

	N	Minimu	Maximu	Mean	Std.
		m	m		Deviation
Supervision	125	2	3	2.86	.349
Benefits and rewards	125	1	4	2.96	.514
Co-workers	125	1	4	2.99	.528
Nature of work	125	1	4	3.09	.509
Promotion	125	2	4	3.15	.443
Communication	125	2	4	2.94	.359
Turnover intention	125	2	4	2.96	.374
Valid N (listwise)	125				

The mean value of supervision, rewards, co-workers, nature of work, promotion and communication are 2.86, 2.96, 2.99, 3.09, 3.15 and 2.94 respectively. The standard deviation for supervision is 3.49%% showing little control of supervision from its mean and the maximum and minimum values for supervision are 2 and 3, respectively.

The standard deviation for communication is 3.59% showing again little dispersion of good communication to employee turnover intention from its mean and the maximum and minimum values of communication are 2 and 4, respectively.

The mean value of fast chance of promotion is 3.15; this indicates that employee turnover is highly tied up with promotion. The standard deviation for promotion is 44.3% which implies higher dispersion from its mean. The maximum value for promotion is 4 and the minimum is 2. This indicates that there is variation on promotion within the bank on a variety of employees' promotion profile.

The mean value for turnover intention is 2.96 and the maximum and minimum values 4 and 2, respectively.

The mean of the mean computed from all those variables is 2.99 which explain that the variables mean is more than average value so they described the dependent variable as it is required.

4.4. Results for Pearson Correlation Coefficient

Table 4.2 depicts the Pearson Correlation for the variables used in the regression model. This analysis is performed to find the relationship between employee turnover intention and

supervisory help, benefits and rewards, feeling for co-workers, nature of work, chance of promotion and good communication.

Table 4.4.1 correlation matrix

Correlations

		ļ ļ	Benefits		Nature		,
	ļ.	!	and	Co-	of		Communication .
		Supervision	rewards	workers	work	promotion	1
Supervision	Pearson	1	.014	010	.092	085	.167**
	Correlation		<u> </u>				
	Sig. (1-		.415	.440	.079	.097	.005
	tailed)						
	N	125	125	125	125	125	125
Benefits and	Pearson	.014	1	.030	.047	123*	.032
rewards	Correlation						
	Sig. (1-tailed)	.415		.326	.238	.031	.315
	N	125	125	125	125	125	125
Co-workers	Pearson	010	.030	1	028	102	.064
	Correlation	!					
	Sig. (1-	.440	.326		.336	.060	.167
	tailed)						
	N	234	234	234	234	234	234
Nature of work	Pearson	.092	.047	028	1	019	.030
	Correlation						
	Sig. (1-	.079	.238	.336		.387	.323
	tailed)						
	N	125	125	125	125	125	125
promotion	Pearson	085	123*	102	019	1	.114*
	Correlation						
	Sig. (1-	.097	.031	.060	.387		.040
	tailed)						
	N	125	125	125	125	125	125
Commission	Pearson	.167**	.032	.064	.030	.114*	1
Communication	Correlation						
	Sig. (1-	.005	.315	.167	.323	.040	
	tailed)						
	N	125	125	125	125	125	125
Turnover	Pearson	.287**	371**	.280**	378**	190**	.333**
intention	Correlation	!					
	Sig. (1-	.000	.000	.000	.000	.002	.000
	tailed)	!					
	N	125	125	125	125	125	125

- **. Correlation is significant at the 0.01 level (1-tailed).
- *. Correlation is significant at the 0.05 level (1-tailed).

In the above correlation matrix the correlation coefficient between the dependent and independent variables shows promotion, benefits and rewards and nature of work are negatively correlated with turnover intention with value of .371,.378 and .190 respectively. On the other supervision, co-workers and communication are positively correlated with the turnover intention.

4.5. Results of the regression analysis

The preliminary analysis made in previous section showed that the assumptions were not violated. In addition when evaluating the significance of the model in ANOVA and Model Summery, the result in the ANOVA for linear regression predicts the outcome variable significant with P-value of 0.000 showing the model applied was significant enough in predicting the outcome variable.

The model summery table also shows that the dependent variable employee turnover intention is 54.2 percent explained by the independent variables. Thus, the regression model used for the study confirmed that important variables that affect the result of the study are included in the model.

Table 4.5.1. Variables Entered/Removed^a

Mode	Variables Entered	Variables	Method
1		Removed	
1	Supervision Benefits and rewards Co-workers Nature of work Promotion Communication Turnover intention		Enter

a. Dependent Variable: Turnover intention

b. All requested variables entered.

Table 4.5.2.Model Summary^b

Mode	R	R Square	Adjusted R	Std. Error of
1			Square	the Estimate

1	.736a	.542	.530	.257
---	-------	------	------	------

a. Predictors: (Constant), supervision, benefits and rewards, co-workers, nature of work, promotion and communication

b. Dependent Variable: Turnover intention

Table 4.5.3.ANOVA^a

Model		Sum of Squares	df	Mean Square	F	Sig.
1	Regression	17.693	6	2.949	44.743	.000b
	Residual	14.961	227	.066		
	Total	32.654	233			

a. Dependent Variable: Turnover intention

b. Predictors: (Constant), supervision, , benefits and rewards, co-workers, nature of work, promotion communication

The regression analysis is conducted using employee turnover intention factors, supervision, benefits and rewards, co-workers, nature of work, promotion communication and the dependent variable turnover intention measured by employee satisfaction with salary increase and chance of promotion.

Tabl	e 4.5.4.Coefficients	ı						
		Unstan	dardized	Standardiz ed Coefficien			95.0 Confid	
Mode	el	Coef	ficients	ts	t	Sig.	Interval	for B
			Std.				Lower	Upper Boun
		В	Error	Beta			Bound	d
1	(Constant)	-1.356	.276		-4.915	.000	-1.900	812
	Supervision	.256	.049	.538	5.178	.000	.158	.353
	Benefit and rewards	.268	.033	668	8.111	.041	.203	.333
	Co-workers	.208	.032	.594	6.481	.000	.145	.272
	Nature of work	.254	.033	.545	7.639	.002	.189	.320
	Promotion	.225	.039	666	5.775	.032	.148	.302

	Communication	.231	.048	.522	4.802	.000	.136	.326
a. Dep	endent Variable: Tu	ırnover ir	ntention					

The result of the multiple regressions for all independent variable indicated that supervision (.538) and co-workers (.594), nature of work (.545) and communication(.522) had positive coefficient(standardized) whereas, benefits and rewards (-.668), and promotion (-.666) had negative coefficient. In addition the P-value for all variables is blow 0.05.

When evaluating the independent variables in the standardized coefficients column the beta value, benefits and rewards made the strongest unique contribution in explaining the dependent variable turnover and the variable that made less contribution is communication within the bank. The result also revealed that all independent variables made a statistically significant contribution to the prediction of the dependent variable ROA.

4.6. Discussion

The above findings suggest that age, gender, promotion opportunities and relationship with coworkers are unimportant in predicting employee turnover in NIB. If an individual has been in one job for a long time, he or she does not need to change his or her job. Supervision mattered a lot in the Bank that it has a high negative relationship with turnover intention. Dissatisfaction with benefit and rewards were the other important factors which lead to turnover intention. Similarly, dissatisfaction with nature of work was the most important predictor of turnover intention in the study. The other important predictor of turnover intention was poor organizational and job specific communication.

Based on the above analysis the next section explained these major results by relating it with the actual practices of the organization in the section that follows.

4.6.1. Supervision support

Supervision support is one of the most important factor in the Bank that predicts turnover intention. A poor relationship with one's immediate superior has been found to be decisive for the decision to leave (Samantrai, 1992). Managers and supervisors influence employees' satisfaction through management styles. A supervisor has the ability to ensure that all employees are satisfied. Turning dissatisfied employees into happy workers requires constant communication, a personable attitude, and the ability to empower and motivate.

Debrah (1993) also noted that a supervisor with poor interpersonal skills and who is also inflexible very quickly drives employees away. The researcher emphasizes the critical role of supervision in retaining employees in NIB.

4.6.2 Benefits and Rewards

Benefits can demonstrate to employees that an organization is supportive and fair, and there is evidence to suggest that stable benefits are at the top of the list of reasons why employees choose to stay with their employer or to join the organization in the first place (Tompkins and Beech, 2002). Employee benefits are constantly evolving as the workforce itself evolves, and as people identify new priorities as being important. It is important to note that the relative importance of benefits will vary according to the specific needs of each individual. (Lochhead C. & Stephens A. 2004). The importance of a benefit plan to an employee with dependents may assume far greater importance particularly in an environment where benefits costs are increasing rapidly (Tompkins and Beech, 2002).

The findings of this study show that employees of NIB seem to be unsatisfied by the benefits and rewards system of the Bank, because the organization doesn't have adequate reward system when compared with other banks.

4.6.3 Nature of the work

Although a widely accepted definition of job satisfaction has not yet been established, job satisfaction generally describes the degree to which somebody is pleased or not with his/her job. Locke (1976, p. 1300) defines job satisfaction as "pleasurable or positive emotional state resulting from the appraisal of one's job or job experiences". According to Spector (1997), job satisfaction is an emotional reaction to work and the different facets of work nature. Some jobs are intrinsically more attractive than others. A job's attractiveness will be affected by many characteristics, including its repetitiveness, challenges, danger, perceived importance and capacity to elicit a sense of accomplishment.

Employees prefer work that is mentally challenging in that it provides them with opportunities to use their skills and abilities and offers a variety of tasks, freedom and feedback on how well they are doing (Robbins 1993). The result of this study is coherent with other available studies in the area as Koh and Goh (1995) found that satisfaction with the nature of work was negatively associated with turnover intention in their sample of clerical employees in the banking industry.

Moreover, satisfaction with the nature of work was found by Westlund and Hannon (2008) to be significantly related to an employee's intention to remain/leave with an organization.

The nature of work in NIB seems more of a routine in most of the positions, specially the tailors; and it is expected that there could be a work load in some workers due to their work assignment and the sensitive nature of the work for fraud can be attributed to the dissatisfaction of employees observed in this aspect.

4.6.4 Organization and Job Specific Communication

Communication plays an enormous part in retaining employees. Without communications, many of practices would be difficult to implement, or it would at least be difficult to implement them with the goal of retaining employees. Organizational communication is the specific process through which information moves and is exchanged throughout the organization. It is through a direct and open communication system that employees understand what is going on in their organization and important organizational behaviors such as organizational citizenship behavior and ethical workforce can be achieved. Due to the nature of the job, several regulations and procedures are put in place regularly as the need arises and because employees who used these procedures into practice are not properly communicated of the change, different employees handle similar cases differently and this has been what customers usually complain about the organization.

5. Summary of Findings, Conclusion and Recommendation

5.1. Summary of Findings

The issue of employee turnover intention is an existing issue which is confronted by several organizations as a result of company's different internal factors and employees own reasons. The researcher has summarized the most important findings in relation to the raised research questions of the study. According to the findings, the major determinant factors for turnover intention in Nib International Bank are lack of fair pay, dissatisfaction with benefits and rewards, lack of work appreciation and lack of salary increment satisfaction followed by poor organizational and job specific communication. Also, the turnover intention is the consequence of dissatisfaction with the nature of the work and poor supervisory support.

There is no good communication channel or medium within the bank to distribute same information at a time for different staffs at various locations.

The time given for various tasks in numerous work units is not enough to complete as per available resources and employees capability to do as it is expected from them.

Most of the responsibilities and tasks given for employees are clear and easily understandable, which employees can do them without spending too long time and resources.

Most employees have no good feelings for their colleagues within the bank on their work assumptions and on how they perform their activities.

Employees are afraid of their supervisors negative thoughts directed to them on numerous operational doings of their tasks.

Various policies and procedures of the bank are not distributed in a wide range of distribution, most of employees fail to response for the effectiveness of the bank's operational policies and procedures.

Supervisors within various posts of the bank are failed to build a good image of themselves within the mind of their subordinates which resulted in that most employees have no good feelings for their supervisors.

Most employees are suffering from challenges for easy communication with their supervisors.

The bank has lack of competent supervisors in various work units as per the view of the respondents.

There is lack of appreciation for good work done by employees, which implies that employees are doing their job without expecting some recognition and appreciations from their supervisors.

There is less remuneration or fair pay system in the bank; employees are dissatisfied with current pay structure of NIB.

Most of the bank's goals are not clear for greatest number of employees in the bank.

Most of workers of the bank are looking for new job.

5.2 Conclusion

The study examined the impact of employee turnover variables such as fair pay, promotion, supervision, operating procedures, rewards, communication, the work itself and relationships with co-workers on employees' turnover intention in Nib International Bank.

As stated in the summary of the findings, the study indicates that employees at Nib International Bank are mostly dissatisfied withtheir salary, rewards andbenefits system, the poor job and organizational specific communication, with the nature of the work and with poor supervisory support. They however, indicated that they are moderately satisfied their work freedom to family time and organizational goals and they are clear with the objectives, in addition they have agreed moderately on their feelings to subordinates.

In terms of the stated research hypotheses the following specific empirical findings emerged from the investigation:

The data collected from the data sources based on the acceptable measurements. The data was presented using descriptive statistics. Correlation and Regression analysis for turnover intention were conducted.

- This study revealed that there does not exist a significant difference among the five age groups of respondents (21 or younger, 22-30 year, 31-40 years, 41-50 years and 51 or older) in their intention to quit their job in NIB.
- The study examined that there exists a significant difference between employees whose working experience is less than 5 years and that of the employees whose working experience is greater than 5 years in their intention to leave NIB.
- The study showed that Gender does not affect turnover intention of respondents (i.e. there is no significant difference between male and female respondents in their intention to leave NIB
- Chance of promotion and proportional rewards within the same industry are slight in Nib which can lead the employees of the bank to shift to another financial institution.

- The study revealed that most employees are looking for another job for the coming years.
- The study discovered that there exists a very significant (positive) relationship between poor communication with NIB and turnover intention.
- The study provides that there is significant difference between employees who are semiprofessionals and those who belong to the senior level management in their intention to leave NIB.

5.3 Recommendation

The researcher suggests the following recommendations based on the findings of the research.

- The management should create a stimulating work environment that makes effective use of people's skills and knowledge, allow employees a degree of autonomy on the job, provide an avenue for them to contribute ideas, and allow them to see how their own contribution influence the organization's well-being.
- The senior managers should encourage the supervisors to be more active in promoting themselves as major source of support in the organization. This can be done by having regular meetings with their subordinates to inquire what forms of practical support may be provided from the organization to help individuals perform better in their job.
- ➤ The bank has to evaluate and re-structure its current salary, benefit and reward packages to refrain its employees from leaving the organization.
- As the bank is increasing its area coverage within the country, existing staffs have to get chance of promotion and entitled with new job opportunities.
- Employees should be communicated on how they have performed in terms of results and goal achievements. Feedback generates change in job behavior if it is related to recognition and reward for exemplary performance.
- > Training should be given to supervisors and other HR managers how to retain employees like.

REFERENCES

- Abbasi, S.M. and Hollman, K.W. (2000). <u>Turnover: the real bottom line.</u> Public Personnel Management: University of Chicago: press Vol. 29, pp. 333-42.
- Abdul Rahman, S.M.M., Naqvi, R. and Ramay, M. (2008). Measuring Turnover intention: A study of IT Professionals in Pakistan. <u>International Review of Business Research papers</u>, 4(3): 45-55.
- Al-Ahmadi, H.A. (2002). "Job satisfaction of nurses in Ministry of Health Hospitals in Riyadh, Saudi Arabia", <u>Saudi Medical Journal</u>, 23, 645-50.
- Ali Shah, I., Fakhr, Z. Ahmad, M. and Zaman K. (2010). Measuring Push, Pull and Personal Factors Affecting Turnover Intention: A case of University Teachers in Pakistan. Review of Economic and Business Studies, 3: 167-192.
- Armstrong, M. (2007). <u>A Handbook of Employee Reward Management and Practice</u>. London and Philadelphia: Kogan Page.
- Arnold, H. J. and Feldman, D. C. (1982). "A Multivariate Analysis of the Determinants of Job Turnover". Journal of Applied Psychology, 67: 350-360.
- Bateman, H. and McAdam, K. (2006). <u>Dictionary of Human Resources and Personnel Management (3rd ed).</u> London: A & C Black Publishers Ltd.
- Bohlander, G., Snell, S and Sherman, A. (2001). <u>Managing Human Resources (12th ed).</u> Cincinnati: OH: South-Western College Publishing.
- Boles J.S., Johnston M.W., and Hair J.F. Jr. (1997. Role stress, work-family conflict, and emotional exhaustion: Interrelationships and effects on some work-related consequences. Journal of Personal Selling and Sales Management, Vol. 17, No. 1, pp.17-28.
- Bowen, D., & Siehl, C. (1997). <u>The future of human resource management: March and Simon (1958) Revisited. *Human Resource Management* (36), ss. Pp. 57-63Business and Management, pp. 551-562. Puket-Thailand.</u>
- Campion, M.A. (1991). Meaning and Measurement in Turnover: Comparison of Alternative Measures and Recommendations for Research. <u>Journal of Applied</u> Psychology, 76: 199-212.
- Carsten, J.M. and SPECTOR, P.E. (1987). Unemployment, Job Satisfaction, and Employee Turnover: A Meta-Analytic Test of the Machesney Model. <u>Journal of Applied Psychology</u>, 72: 374-381.
- Cornog G. Y.(1957). <u>The Personnel turnover concept: A reappraisal.</u> Public administration Review: Vol17, No 4 pp247-256
- Cotton, J.L. and Tuttle, J.M. (1986). Employee turnover: a meta-analysis and review with implications for research. Academy of Management Review, Vol. 11, pp. 55-70.
- Duxbury, Linda and Chris Higgins ,(2001). Work-Life Balance in the New Millennium: Where are We? Where do We Need to Go? for the next

- millennium. Journal of Management, 26, 463-488.
- Firth, L., Mellor, D.J., Moore, K.A. and Loquet, C. (2004). How can managers reduce employee intention to quit? <u>Journal of Managerial Psychology</u>, Vol. 19 No. 2, pp. 170-87.
- Geoffrey, Marczyk, David DeMatteo, and David Festinger. (2005). <u>Essentials of Research Design and Methodology.</u> New Jersey: John Wiley & Sons, Inc.
- Gomez-Mejia, L.R., Balkin (2001). <u>Managing Human Resources 3rd ed.</u> Upper SaddleRiver, NJ: Prentice Hall.
- Griffeth R. & Hom P.(2004). <u>Innovative Theory and Empirical Research on</u> EmployeeTurnover. Information Age Publishing
- Griffeth, R. W., Hom, P. W., & Gaertner, S. (2000). <u>A meta-analysis of antecedents and correlates of employee turnover.</u> Update, moderator tests, and research implications
- Hackman, J.R. and Suttle J. L. (1977). <u>Improving Life at Work</u>. Santa Monica, Califonia: Goodyear Publishing Company, INC.
- Hom, P. W., & Griffeth, R. W. (1995). <u>Employee turnover</u>. Cincinnati, OH: South-Western.
- Heery, E. and Noon, M. (2001). <u>A dictionary of human resource management</u>. New York: Oxford University Press.
- Ivancevich John and Glueck William, (1989). Foundation of Personnel /Human Resource Management. IRWN, Boston
- Jenkins S. (1988. Turnover: Correcting the Cause. <u>Personnel Journal</u>, December pp 43-48 John Bratton and Jeffrey Gold (1999). <u>Human Resource Management: theory and practice (</u>2nd ed. Macmillan Press Ltd.
- Khan, T. and Mishra G.P.(2013). Promotion as Job Satisfaction, A study On Colleges of Muscat, Sultanate of Oman. <u>European Journal of Business and Management</u>, 5(5): 56-62.
- Koh, H. and Goh, C. (1995). An Analysis of the Factors Affecting the Turnover Intention of Non Managerial Clerical Staff: A Singapore Study. <u>The International Journal of Human Resource Management</u>, 6(1).
- Kondalkar, V.G. (2007). <u>Organizational Behavior</u>. New Delhi: New Age International (P) Limited Publishers.
- Kottke, J. and Sharafinski, C. (1988). <u>Measuring perceived supervisory and organizational support</u>. Educational and Psychological Measurement: Vol. 48 No. 4,pp. 1075-9
- Lee, T. W., Mitchell, T. R. (1994). <u>An Alternative Approach: The Unfolding Model</u>
 <u>OfVoluntary Employee Turnover</u>. The Academy of Management Review:vol.19, 51-89,
 Vol. 19, No 1. URL: http://www.jstor.org/stable/258835
- Lee, T. W., Mitchell, et al. (1999). The unfolding model of voluntary turnover: A replication and extension. <u>Academy of Management Journal</u>, *42*,
- Lee, T. W., Mitchell, T. R., Sablynski, C. J., Burton, J. P., & Holtom, B. C. (October2004). The Effects of Job Embeddedness on Organizational Citizenship, JobPerformance, Volitional Absences, and Voluntary Turnover. Academy of Management vol. 47, No:5, pp. 711-722.
- Lochhead C.& Stephens A. (2004). Employee Retention ,Labour Turnover &

- Knowledge Transfer. Case Studies from the Canadian Plastics Sector
- Loquercio, David, Hammersley Mark and Emmens, Ben, (2006). Understanding and Addressing Staff Turnover in Humanitarian Agencies.
- Lucas G. H. Jr; A. Parasuraman, Davis R. A. & Enis B. M. (1984). An Empirical Study of Ramlall, S. (2004). A Review of Employee Motivation Theories and theirImplications for Employee Retention within Organizations. <u>Journal of AmericanAcademy of Business</u>.
- Maertz, C. Jr., Griffeth, R., Campbell, N. and Allen, D. (2007). The effects of Perceived organizational support and perceived supervisor support on employee turnover. <u>Journal of Organizational Behavior</u>, Vol. 28 No. 8, pp. 1059-75.
- Martin C.(2003). Explaining Labour Turnover. <u>Empirical Evidence from UK</u> <u>Establishments</u>, Labour Vol 17 No 3
- McPherson, B.D. (1976). <u>Involuntary turnover and organizational effectiveness in TheNational Hockey League, in Gruneau, R.S. and Albinson, J.G. (Eds).</u> Canadian Sport: Sociological Perspectives, Addison-Wesley, Don Mills, pp. 259-75.
- Meyer, John, Laryssa Topolnytsky, Henryk Krajewski and Ian Gellatly(2003). <u>Best Practices: Employee Retention</u>. Toronto: Tomson-Carswell
- Mobley, W. H. (1977). <u>Intermediate linkages in the relationship between job satisfaction</u> and employee turnover. Journal of Applied Psychology, *62*, 237–240
- Mobley, W.H.(1982). <u>Employee turnover: causes, consequences, and control</u>. Addison Wesley Publishing Company Inc.
- Ng, T. and Sorensen, K. (2008). <u>Toward a further understanding of the relationships</u> between perceptions of support and work attitudes: a meta-analysis. Group and Organization Management, Vol. 33 No. 3, pp. 243-68.
- Price, J.L. (1989). The impact of turnover on the organization. Work and Occupations Vol. 16,
- Robbins, S.P. (1997). <u>Essentials of Organizational Behavior.</u> New Jersey: Upper Saddle River
- Ronra, B., & Chaisawat, M. (2009). Factors Affecting Employee Turnover and Job Satisfaction. <u>A Case Study of Amari Hotels and Resorts</u>.
- Singh BP, Chhabra T and Taneja P,(1994). <u>Personnel Management and Industrial</u>
 <u>Relations.</u> Dhanpat Rai and Sons, Delhi Stan Kossen, The Human Side of Organization,
 5th ed. Harper Collins, New York, 1991
- Spector, P. E. (1997). <u>Job Satisfaction: Application, Assessment, Cause and Consequences</u>. California: Sage Publications.
- Spector, P.E. (2008). <u>Industrial and Organizational Behavior</u>. Oxford, England: Wiley.
- Taylor, Stephen (1998). <u>Employee Resourcing.</u> Cromwell Press, Wiltshire, 1998, pp 238-241
- Teeraprasert, S., Piriyakul, R., & Khantanapha, N. (2012). <u>Influencing Factors to</u>
 <u>Job Satisfaction and Organizational Commitment in Manufacturing Organization, Affect to Turnover Intention.</u> A Context of Thai Fashion Industr: International Conference onBusiness and Management, pp. 551-562. Puket- Thailand.
- Williams C. R.; Livingstone L. P. (1994). Another Look at the Relationship

between Performance and Voluntary Turnover. <u>The Academy of Management Journal</u>, Vol. 37 No 2, pp 269-298

William H. Mobley(1982). Employee Turnover: Causes, Consequences, and Control. Addison-Wesley Publishing Company Inc.

Withers, Pam. (2001). The six secrets to attracting and retaining great employees. <u>CMAManagement</u> 75(7); 24–26.

ELECTRONIC REFERENSES

www.acas.org.uk www.amanet.org/books/catalog/0814405975_ch.htm www.agr.com/Our-Services/Field www.cipd.co.uk/surveys www.wisegeek.com/what-is employee-turnover.htm

APPENDEX

St. MARY'S UNIVERSITY SCHOOL OF GRADUATE STUDIES MASTER OF BUSINESS ADMINISTRATION IN GENERAL MANAGEMENT

TOPIC- FACTORS AFFECTING EMPLOYEE TURNOVER NTENTION IN CASE OF NIB INTERNATIONAL BANK S.C

Dear Respondent:

The main Objective of this Questionnaire is to gather data and weigh the opinion of employees of Nib International Bank regarding the organizational factors that affect the turnover intention of employees. The data and opinion will be strictly used for writing a research thesis which shall be submitted in partial fulfillment of Masters Degree in Business Administration in General Management.

Thus your honest, immediate and thoughtful response will be appreciated, and certainly make the research work fruitful.

Note:

- Please do not write your name.
- After filling all the questions, insert the questionnaire in the envelope provided together with the questionnaire and seal it and hand it to the researcher.

If you have any question to ask please don't hesitate to contact me at any time through the following address. Tel-.0911 41 97 20

Please put a tick (✓) mark just inside the given box

	base par a rick () mark je	150 11	iside the given box.
1.	Gender		
	☐ Male		Female
2.	Age		
	\square 21 years and younger		\square 41- 50 years old
	□ 22-30 years old		☐ 51 years and older
	\square 31-40 years old		

3. Marital Status

		THE ONE NUMBER FOR EACH MES CLOSEST TO REFLECTING	Strongly disagree	Disagi
		Part Two		
	☐ 12-13 (S	uire to be futeu by Current Employee	s vj 141 D	
	П 0 11 (I av	aire to be filled by Current Employee	es of NIR	
	☐ 6-8 (professional)			
	☐ 1-5 (Semi –professional)			
7.	Your Job Grade in NIB			
	□ 6 to 10 Years	☐ Greater than 20 Years		
	☐ 3 to 5 Years	□11 to 20 Years		
0.	☐ Less than 3 Years			
6.	For how long have you been e	employed in this organization? (Tenure)		
	☐ Customer Service	☐ Back Office ☐Bot	h □Others	
5.	What is the category of your v	work?		
	□ Degree			
	□ Diploma	☐ If any other, please specify		
	☐ Certificate	☐ Doctorate Degree		
	☐ High school Complete	☐ Master's Degree		
4.	Educational Level			
	□Married	☐ Widowed		
	□Single	☐ Divorced		

	PART TWO PLEASE CIRCLE THE ONE NUMBER FOR EACH QUESTION THAT COMES CLOSEST TO REFLECTING YOUR OPINION ABOUT IT.		trongly or Disa	y disagi agree	Dis Neither	agree Agree Agree gly agree
1	I feel I am being paid a fair amount for the work I do.	5	1	2	3	4
2	Raises are too few and far between.	5	1	2	3	4

3	I feel unappreciated by the organization when I think about what they pay me.	5	1	2	3	4	
4	I feel satisfied with my chances for salary increases.	5	1	2	3	4	
5	There is really too little chance for promotion on my job.	5	1	2	3	4	
6	Those who do well on the job stand a fair chance of being promoted.	5	1	2	3	4	
7	People get ahead as fast here as they do in other places.	5	1	2	3	4	
8	I am satisfied with my chances for promotion.	5	1	2	3	4	
9	My supervisor is quite competent in doing his/her job.	5	1	2	3	4	
10	My supervisor is unfair to me.	5	1	2	3	4	
11	My supervisor shows too little interest in the feelings of subordinates.	5	1	2	3	4	
12	My supervisor goes out of her/his way to make my work life easier for me.	5	1	2	3	4	
13	It is easy for me to talk with my immediate supervisor about my job related problem.	5	1	2	3	4	
14	I relied on my immediate supervisor when things get tough at my job	5	1	2	3	4	
15	I like my supervisor.	5	1	2	3	4	
16	I am not satisfied with the benefits I receive.	5	1	2	3	4	
17	The benefits we receive are as good as most other organizations offer.	5	1	2	3	4	
18	The benefit package we have is equitable.	5	1	2	3	4	
19	There are benefits we do not have which we should have.	5	1	2	3	4	
20	When I do a good job, I receive the recognition for it that I should receive.		1	2	3	4	5
21	I do not feel that the work I do is appreciated.		1	2	3	4	5
22	There are few rewards for those who work here.		1	2	3	4	5
23	I don't feel my efforts are rewarded the way they should be.		1	2	3	4	5
24	Many of our rules and procedures make doing a good job difficult.		1	2	3	4	5

l		1					
25	My efforts to do a good job are seldom blocked by red tape.	1	2	3	4	5	
26	I have too much to do at work.	1	2	3	4	5	
27	I have too much paperwork.	1	2	3	4	5	
28	I like the people I work with.	1	2	3	4	5	
29	I find I have to work harder at my job because of the incompetence of people I work with.	1	2	3	4	5	
30	I enjoy my coworkers.	1	2	3	4	5	
31	There is too much bickering and fighting at work.	1	2	3	4	5	
32	My job responsibilities are clear to me	1	2	3	4	5	
33	I like doing the things I do at work.	1	2	3	4	5	
34	I feel a sense of pride in doing my job.	1	2	3	4	5	
35	It is clear to me what others expect of me at my job.	1	2	3	4	5	
36	I am given enough time to do what is expected of me at my job.	1	2	3	4	5	
	PLEASE CIRCLE THE ONE NUMBER FOR EACH QUESTION THAT COMES CLOSEST TO REFLECTING YOUR OPINION ABOUT IT.		Strongly disagree Disagree Neither Agree nor Disagree Agree Strongly agree				
			ly agree				
37	I feel frustrated at my job.		ly agree	3	4	5	
37 38	I feel frustrated at my job. My job gives me enough time for family activities.	Strong		3 3	4 4	5 5	
		Strong 1	2				
38	My job gives me enough time for family activities.	Strong 1	2 2	3	4	5	
38	My job gives me enough time for family activities. Communications seem good within this organization.	Strong 1 1	2 2 2	3	4	5	
38 39 40	My job gives me enough time for family activities. Communications seem good within this organization. The goals of this organization are not clear to me. I often feel that I do not know what is going on with the	Strong 1 1 1	2 2 2 2	3 3 3	4 4	5 5 5	
38 39 40 41	My job gives me enough time for family activities. Communications seem good within this organization. The goals of this organization are not clear to me. I often feel that I do not know what is going on with the organization.	Strong 1 1 1 1 1	2 2 2 2 2	3 3 3 3	4 4 4	5 5 5 5	

45	I will probably look for a new job in the next year	1	2	3	4	5

Thank You Very Much for your Cooperation!